Author's response to reviews

**Title:** The influence of long-term treadmill exercise on bone mass and articular cartilage in ovariectomized rats

**Authors:**

- Ting-Kuo Chang (tomy4367@gmail.com)
- Chang-Hung Huang (changhung0812@gmail.com)
- Chun-Hsiung Huang (chhuang1230@gmail.com)
- Hsuan-Chiang Chen (mmh.b918@ms1.mnh.org.tw)
- Cheng-Kung Cheng (ckcheng@ym.edu.tw)

**Version:** 2  **Date:** 25 June 2010

**Author's response to reviews:** see over
Dear Dr. Norton:

Thank you for your letter dated on June 8, 2010, which included a comprehensive review of our manuscript entitled “The influence of long-term treadmill exercise on bone mass and articular cartilage in ovariectomized rats” by you and your referee panel. According to the reviewers’ comments, my colleague and I had revised the manuscript with underline in the text. All co-authors have read and agreed with the contents of the manuscript and this manuscript has not been submitted or published elsewhere. Enclosed please find the revised manuscript and a point by point response list to editorials and reviewers.

Thank you again for your kind review.

Sincerely Yours,

Cheng-Kung Cheng, Prof. Ph.D.
Institute of Biomedical Engineering
National Yang Ming University
No. 155, Sec. 2, Li-Nung St.,
Shih-Pai, Taipei 11221,
TAIWAN
Co-authors:

Ting-Kuo Chang  
Chang-Hung Huang  
Chun-Hsiung Huang  
Hsuan-Chiang Chen